Author's response to reviews

Title: Methods of Competing Risks Analysis of End-Stage Renal Disease and Mortality among People with Diabetes

Authors:

Hyun J Lim (hyun.lim@usask.ca)
Xu Zhang (matxxz@langate.gsu.edu)
Roland Dyck (roland.dyck@usask.ca)
Nathaniel Osgood (osgood@cs.usask.ca)

Version: 4 Date: 15 October 2010

Author's response to reviews: see over
Dear Reviewer #2,

We understand your concern on the L-M model. We pointed out the extra assumption of the LM-model and thus limitation of the applicability in the Discussion (page 18).

Thank you.

Hyun J. Lim
Corresponding Author